Author's response to reviews

Title: A comparison of inpatient with outpatient balloon catheter cervical ripening: a pilot randomized controlled trial.

Authors:

Chris S Wilkinson (chrisswilkinson@mac.com)
Pamela Adelson (pamela.adelson@adelaide.edu.au)
Deborah A Turnbull (deborah.turnbull@adelaide.edu.au)

Version: 5 Date: 28 April 2015

Author's response to reviews:

Changes undertaken by the authors in response to the editorial comment

1. Unfortunately the version of the ms that we have does not contain the name of the IRB or the confirmation that consent was written. Please include these in the ms.

   The name of the hospital ethics committee (IRB) has been inserted in line 100.

2. Include the Funding details in the Acknowledgement section.

   The funding details have been merged into the Acknowledgement section (line 417)

3. Include the figure legend after the References section.

   The title and the figure legend have been incorporated into the figure.